Notes of the RICH1 Phone meeting 22/06/2007
People Present:
CERN:  Fabio, Olav
IC:         Bill, Dave, Geoff
Oxford:  Brian, Rohan
Bristol:   Nick, Richard

Apologies for absence:
Neville
Discussed Agenda Topics:
1. Gas Enclosure
The beampipe tests finished June 20th. All heating jackets have been removed. The beampipe is at atmospheric pressure filled with neon. Work around the beampipe is now permitted. The restraints to the G.E. have been fixed (clearance ~0.3 mm) and the shims have been removed. Richard P. is preparing the temperature sensors and will install them in the G.E. 
2. Mirrors
Flat: The lower and upper flat mirrors have been aligned via survey to better than ~0.3 mrad.  The upper mirrors will be mounted onto the rig and their alignment will be checked next Thursday 28th June. The mirrors will be stored in their installation boxes in Bat. 156 until installation in RICH1. The survey reports will be prepared by the survey team during the next two weeks.
Spherical: The mounting and assembly of the CMA mirrors and their frame will begin Monday 2nd July. Geoff will probably come out to CERN on Monday 2nd July to help. The CMA mirrors will be aligned and then monitored to check the D0 and the CoC spot position over a period of ~2 weeks. The lower support rail can not be found anymore. A new one will be manufactured at IC and brought out to CERN.

Installation: 
It is foreseen to install the mirrors towards the end of July. Tim or Richard will come out to CERN to help in case of mechanical problems during installation and to do two related machining jobs (see below) for which Nick requested drawings (action Dave).  First the spherical mirrors will be installed because access into RICH1 is needed to check the mirror-beampipe gap. The CERN transport service has been already contacted and they will transport the mirrors to the LHCb pit.  After installation of the mirrors RICH1 will be filled with dry air.

The bosses on the RICH1 doors need to be trimmed due to the presence of the CMA label on the C-frame and the horizontal rig plates need to be drilled to serve as protection for the CF mirrors after installation.  During transport the flat mirror plate V-blocks need to be clamped to the rail to prevent the plate from falling off.
3. Aerogel
It is risky to install the aerogel soon because it needs to be protected from air. The aerogel will be installed when there is no more need to open RICH1, probably around Dec07. The Milano technicians wish to test the aerogel mechanics. It is preferable to have this done before the installation of the mirrors and with the presence of Delio i.e. ~week 9-13 July.
4. HPD
Column Dressing:  The 6th column of the second batch has been installed in the SSB box and left overnight.  HPD’s are required for the 7th column. A batch of 10 will arrive from Scotland these days.

Rail System: NAB promised to deliver next week. Following assembly tests in Oxford, it will be transported to CERN, mounted and load tests will be undertaken, probably around end July/early August.
Lower Rail: Concept design ready. Brian will send it to IC by Monday.
Mechanics: The drawings are being delivered. The workshop is producing the parts. It is planned to send the chariot to CERN for testing with the upper rail assembly.

Cabling:  The copper cabling for both upper and lower HPDs is installed. The cables for the MDMS, LAMS and optical fibers need to be installed. Ken prefers to delay the installation of the fibers until the HPDs’s are put in. Maria and Vera came to CERN to help with the fixing of connectors (~70% completed).

Light Leak detector:  Richard P. is taking over from Gabriel who has left IC.
Photon Funnel: Ray constructed a model to test the MDMS. This raises questions which need to be discussed.

5. TT
Dave had a meeting in the pit with the TT responsible. The TT sensors will be installed during August and September. They wish to avoid having the detectors open during mechanical operations. Some of the installation can be done after hours to avoid conflicts. The TT sensor installation will first start on the A-side and then move to the C-side.

6. HPD Cooling
The RICH2 cooling is running. There is a difference between the A and C sides which is not yet understood. For RICH1 it is proposed to use only 2 valves:  input and output.
7. LAMS

Received from P. Soler the list of required components.
8. Quartz Windows
D. Morrow is available any month. The glueing of the QW to the support frame will be probably done at the end of summer.
9. Next meeting
Friday 6th July 2007 at 10:00(GMT) 11:00(CET). 
